\documentclass{aims}
\usepackage{amsmath}
  \usepackage{paralist}
  \usepackage{graphics}
\usepackage{epsfig} %For pictures: screened artwork should be set up with an 85 or 100 line screen
 \usepackage[colorlinks=true]{hyperref}
\hypersetup{urlcolor=blue, citecolor=red}
%\usepackage{hyperref}

  \textheight=8.2 true in
   \textwidth=5.0 true in
    \topmargin 30pt
     \setcounter{page}{1}

% The next 5 line will be entered by an editorial staff.
\def\currentvolume{X}
 \def\currentissue{X}
  \def\currentyear{200X}
   \def\currentmonth{XX}
    \def\ppages{X--XX}



\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}{Corollary}
\newtheorem*{main}{Main Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}{Proposition}
\newtheorem{conjecture}{Conjecture}
\newtheorem*{problem}{Problem}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{remark}{Remark}
\newtheorem*{notation}{Notation}
\newcommand{\ep}{\varepsilon}
\newcommand{\eps}[1]{{#1}_{\varepsilon}}
\newcommand{\RR}{\mathbb R}


\title[Noncoercive elliptic equations]
      {Noncoercive elliptic equations with subcritical growth}

\author[Vicen\c tiu D. R\u adulescu]{}


\subjclass{Primary: 35J60; Secondary: 35A15; 58E05; 58E30.}
 \keywords{Nonlinear elliptic equation, Dirichlet boundary condition, Dual variational method, Mountain pass geometry.}


 \email{vicentiu.radulescu@imar.ro; vicentiu.radulescu@math.cnrs.fr}




\thanks{This paper is dedicated with gratitude to Professor Marius Iosifescu on his 75th birthday anniversary.}

\begin{document}
\begin{thebibliography}{99}

\bibitem{amra} (MR0370183)
    \newblock A. Ambrosetti and P. Rabinowitz,
    \newblock \emph{Dual variational methods in critical point theory and applications},
    \newblock  J. Functional Analysis, \textbf{14} (1973), 349--381.

\bibitem{bonmar} (MR1483627)
\newblock G. Bonanno and S.A. Marano,
\newblock \emph{Positive solutions of elliptic equations with discontinuous nonlinearities},
\newblock Topol. Methods Nonlinear Anal., \textbf{8} (1996), 263--273.

\bibitem{brebook} (MR0697382)
    \newblock H. Brezis,
    \newblock ``Analyse Fonctionnelle. Th\'eorie et Applications",
    \newblock  Masson, Paris, 1983.

\bibitem{breams} (MR0693957)
    \newblock H. Brezis,
    \newblock \emph{Periodic solutions of nonlinear vibrating strings and duality principles},
    \newblock  Bull. Amer. Math. Soc., \textbf{8} (1983), 409--426.

\bibitem{brenir} (MR1127041)
    \newblock H. Brezis and L. Nirenberg,
    \newblock \emph{Remarks on finding critical points},
    \newblock  Comm. Pure Appl. Math., \textbf{44} (1991), 939--963.

\bibitem{carlmot} (MR2397751)
\newblock S. Carl and D. Motreanu,
\newblock \emph{Constant-sign and sign-changing solutions for nonlinear eigenvalue problems},
\newblock Nonlinear Anal., \textbf{68} (2008), 2668--2676.

\bibitem{cla1} (MR0534415)
    \newblock F. Clarke,
    \newblock \emph{A classical variational principle for periodic Hamiltonian trajectories},
    \newblock  Proc. Amer. Math. Soc., \textbf{76} (1979), 186--188.

\bibitem{cla2} (MR0614215)
    \newblock F. Clarke,
    \newblock \emph{Periodic solutions to Hamiltonian inclusions},
    \newblock  J. Differential Equations, \textbf{40} (1981), 1--6.

\bibitem{eke} (MR0723579)
    \newblock I. Ekeland,
    \newblock \emph{A perturbation theory near convex Hamiltonian systems},
    \newblock  J. Differential Equations, \textbf{50} (1983), 407--440.

\bibitem{fpr} (MR2498753)
\newblock R. Filippucci, P. Pucci and F. Robert,
\newblock \emph{On a $p$-Laplace equation with multiple critical nonlinearities},
\newblock J. Math. Pures Appl., {\bf 91} (2009), 156--177.

\bibitem{ghpr} (MR1030853)
\newblock N. Ghoussoub and D. Preiss,
\newblock \emph{A general mountain pass principle for locating and classifying critical points},
\newblock Ann. Inst. H. Poincar\'e Anal. Non Lin\'eaire, \textbf{6} (1989), 321--330.

\bibitem{giltru} (MR1814364)
     \newblock D. Gilbarg and N.S. Trudinger,
     \newblock ``Elliptic Partial Differential Equations of Second Order",
     \newblock Reprint of the 1998 edition, Springer-Verlag, Berlin, 2001.

\bibitem{krvbook} (MR2683404)
     \newblock A. Krist\'aly, V. R\u adulescu and Cs. Varga,
     \newblock ``Variational Principles in Mathematical Physics, Geometry, and Economics. Qualitative Analysis of Nonlinear Equations and Unilateral Problems",
     \newblock Encyclopedia of Mathematics and its Applications, vol.~136, Cambridge University Press, Cambridge, 2010.

\bibitem{lanlaz} (MR0267269)
    \newblock E.M. Landesman and A.C. Lazer,
    \newblock \emph{Nonlinear perturbations of linear elliptic boundary value problems at resonance},
    \newblock  J. Math. Mech., \textbf{19} (1969/1070), 609--623.

\bibitem{marmot} (MR1764892)
\newblock S.A. Marano and D. Motreanu,
\newblock \emph{Existence of two nontrivial solutions for a class of elliptic eigenvalue problems},
\newblock Arch. Math. (Basel), \textbf{75} (2000), 53--58.

\bibitem{Pa} (MR0259955)
\newblock R. Palais,
\newblock \emph{Ljusternik-Schnirelmann theory on Banach manifolds},
\newblock Topology, {\bf 5} (1966), 115--132.

\bibitem{palsma} (MR0158411)
\newblock R. Palais and S. Smale,
\newblock \emph{A generalized Morse theory},
\newblock Bull. Amer. Math. Soc., {\bf 70} (1964), 165--171.

\bibitem{puccirad}
\newblock P. Pucci and V. R\u adulescu,
\newblock \emph{The impact of the mountain pass theory in nonlinear analysis: a mathematical survey},
\newblock Boll. Unione Mat. Ital., \textbf{IX} (2010), 543--584.

    \bibitem{PS1} (MR0766489)
    \newblock P. Pucci and J. Serrin,
    \newblock \emph{Extensions of the mountain pass theorem},
    \newblock J. Funct. Anal., {\bf 59} (1984), 185--210.

\bibitem{PS2} (MR0808262)
\newblock P. Pucci and J. Serrin,
\newblock \emph{A mountain pass theorem},
\newblock J. Differential Equations, {\bf 60} (1985), 142--149.

\bibitem{PS3} (MR0869402)
\newblock P. Pucci and J. Serrin,
\newblock \emph{The structure of the critical set in the mountain pass theorem},
\newblock Trans. Amer. Math. Soc., {\bf 299} (1987), 115--132.

\bibitem{radhindawi} (MR2572778)
     \newblock V. R\u adulescu,
     \newblock ``Qualitative Analysis of Nonlinear Elliptic Partial Differential Equations: Monotonicity, Analytic, and Variational Methods",
     \newblock Contemporary Mathematics and Its Applications, vol.~6, Hindawi Publishing Corporation, New York, 2008.

    \bibitem{radcluj}
     \newblock V. R\u adulescu,
     \newblock \emph{Remarks on a limiting case in the treatment of nonlinear problems with mountain pass geometry},
     \newblock Universitatis Babes--Bolyai Mathematica, \textbf{LV}, No. 4 (2010), 99--106.

\bibitem{toland} (MR0522706)
     \newblock J. Toland,
     \newblock \emph{A duality principle for nonconvex optimisation and the calculus of variations},
     \newblock Arch. Rational Mech. Anal., \textbf{71} (1979), 41--61.

\bibitem{zheng} (MR1051229)
     \newblock X.M. Zheng,
     \newblock \emph{Un r\'esultat de non-existence de solution positive pour une \'equation elliptique},
     \newblock Ann. Inst. H. Poincar\'e Anal. Non Lin\'eaire, \textbf{7} (1990), 91--96.

\end{thebibliography}

\maketitle


\centerline{\scshape Vicen\c tiu D. R\u adulescu }
\medskip
{\footnotesize
 \centerline{``Simion Stoilow" Mathematics Institute of the
Romanian Academy}
   \centerline{P.O. Box 1-764, 014700 Bucharest, Romania}
   \centerline{ Department of Mathematics, University of Craiova, 200585 Craiova, Romania}
}

\bigskip

 \centerline{(Communicated by Salvatore Angelo Marano)}


\begin{abstract}
We study a class of nonlinear elliptic equations with subcritical growth and Dirichlet boundary condition. Our purpose in the present paper is threefold: (i) to establish the effect of a small perturbation in a nonlinear coercive problem; (ii) to study a Dirichlet elliptic problem with lack of coercivity; and (iii) to consider the case of a  monotone nonlinear term with subcritical growth. This last feature enables us to use a dual variational method introduced by Clarke and Ekeland in the framework of Hamiltonian systems associated with a convex Hamiltonian and applied by Brezis to the qualitative analysis of large classes of nonlinear partial differential equations. Connections with the mountain pass theorem are also made in the present paper.
\end{abstract}


\section{Introduction}

Let $\Omega\subset\RR^N$ be a bounded domain with smooth boundary. Let $2^*$ denote the critical Sobolev exponent, that is, $2^*=2N/(N-2)$ if $N\geq 3$ and $2^*=+\infty$ if $N\in\{1,2\}$. Throughout this paper we denote by $\lambda_1$ the first eigenvalue of the Laplace operator $(-\Delta)$ in $H^1_0(\Omega)$.

In this paper we are concerned with the nonlinear elliptic problem
\begin{equation}\label{p}\left\{
\begin{array}{lll}
&-\Delta u-\lambda u = f(u)\qquad &\mbox{in $\Omega$}\\
 &u=0\qquad &\mbox{on $\partial\Omega$},
\end{array} \right.
\end{equation}
where $\lambda$ is a real parameter. The function $f\in C^1(\RR,\RR)$ is assumed to satisfy the following hypotheses:
\begin{equation}\label{f1}
f(0)=f'(0)=0;
\end{equation}
\begin{equation}\label{f2}
|f(u)|\leq C(1+|u|^p)\qquad\mbox{for all $u\in\RR$, where $1<p<2^*-1$};\end{equation}
and there exists $\mu>2$ such that for all $u>0$ large enough,
\begin{equation}\label{f3}
0<\mu F(u)\leq uf(u),\qquad\mbox{where $F(u):=\int_0^uf(t)dt$.}\end{equation}

Under these assumptions, Ambrosetti and Rabinowitz \cite{amra} proved that problem (\ref{p}) has at least one {\it positive}  solution, provided that $\lambda<\lambda_1$.
We are looking for weak solutions of (\ref{p}), that is, $u\in H^1_0(\Omega)$ with $u>0$ a.e. in $\Omega$ and such that for all $\xi\in H^1_0(\Omega)$,
$$\int_\Omega (\nabla u\cdot\nabla \xi-\lambda u\xi)dx=\int_\Omega f(u)\xi dx\,.$$
A standard bootstrap argument (see Gilbarg and Trudinger \cite{giltru}) that combines Schauder and H\"older regularity shows that if $u$ is a weak solution then $u$ is a classical solution of problem (\ref{p}). We refer to R\u adulescu \cite[pp.~4-5]{radhindawi} for complete details of such a regularity argument.

Returning to problem (\ref{p}), we point out that a solution of this problem exists provided that the linear operator $-\Delta -\lambda I$ is {\it coercive} in $H^1_0(\Omega)$, that is, there exists $c>0$ such that for all $\xi\in H^1_0(\Omega)$,
$$\int_\Omega (|\nabla \xi|^2-\lambda \xi^2)dx\geq c\,\|\xi\|^2_{H^1_0}.$$
The hypothesis $\lambda<\lambda_1$ corresponds to the existence of a mountain pass geometry near the origin for the energy functional associated to problem (\ref{p}).

The same argument based on the mountain pass theorem can be extended to the case where $f$ has an {\it almost critical growth}, more precisely if assumption (\ref{f2}) is replaced with
$$\lim_{u\rightarrow +\infty}\frac{f(u)}{u^{(N+2)/(N-2)}}=0,$$
provided that $N\geq 3$.

We observe that the {\it technical} assumption (\ref{f3}) implies a super-linear behaviour of the nonlinear term. Indeed, a straightforward computation shows that, by (\ref{f3}), there are $C_1,\, C_2>0$ such that
\begin{equation}\label{f4}
f(u)\geq C_1u^{\mu-1}-C_2,\qquad\mbox{for all $u\geq 0$}.\end{equation}
To the best of our knowledge, it is not known whether the above existence result for problem (\ref{p}) still remains true if assumption (\ref{f3}) is replaced under the weaker hypothesis
(\ref{f4}).

In the limiting case  $p=1$ in hypothesis (\ref{f2}) that corresponds to a linear growth of $f$, we have provided in \cite{radcluj} a sufficient condition for the existence of a nontrivial solution to problem (\ref{p}). This corresponds to a weak slope of $f$ near the origin, combined with a faster (linear) growth of $f$ in a neighbourhood of $+\infty$.

We refer to the seminal paper by Brezis and Nirenberg \cite{brenir} for several related existence and nonexistence results.

\section{The main results}
In the present paper we have two objectives.
We are first interested in the effect of a certain {\it perturbation} in problem (\ref{p}). More precisely, we consider the problem
\begin{equation}\label{p1}\left\{
\begin{array}{lll}
&-\Delta u-\lambda u = f(u)+h(x)\qquad &\mbox{in $\Omega$}\\
&u>0\qquad &\mbox{in $\Omega$}\\
 &u=0\qquad &\mbox{on $\partial\Omega$},
\end{array} \right.
\end{equation}
where $h\in L^\infty(\Omega)$. We observe that if the perturbation $h$ is {\it big} with respect to a suitable topology, then problem (\ref{p1}) does not have any solution. Indeed, let $\phi_1$ be a positive eigenfunction of $(-\Delta)$ corresponding to $\lambda_1$. By multiplication in (\ref{p1}) with $\phi_1$ and integration by parts we obtain
$$\int_\Omega h\phi_1dx=\int_\Omega\left[(\lambda_1-\lambda)u-f(u)\right]\phi_1dx.$$
Thus, by (\ref{f4}),
\begin{equation}\label{phi1}\int_\Omega h\phi_1dx\leq\int_\Omega\left[(\lambda_1-\lambda)u-C_1u^{\mu-1}+C_2\right]\phi_1dx\leq
M\int_\Omega\phi_1dx,\end{equation}
where $$M:=\max\{(\lambda_1-\lambda)t-C_1t^{\mu-1}+C_2;\ t\in [0,\infty)\}.$$
Since $\mu>2$ and $\lambda<\lambda_1$, it follows that $M\in (0,\infty)$. Returning to relation (\ref{phi1}), we conclude that problem (\ref{p1}) cannot have any solution if $h$ is {\it nonnegative} and $\|h\|_{L^\infty}$ is large enough. We refer to Zheng \cite{zheng} for a related nonexistence result in the case where $h$ is {\it negative} and $\|h\|_{L^\infty}$ is sufficiently large.

In contrast, our first main result establishes that if the perturbation is {\it small} with respect to the $L^\infty$--topology, then problem (\ref{p1}) admits solution.

\begin{theorem} \label{t1}
 Assume $\lambda<\lambda_1$ and hypotheses (\ref{f1})--(\ref{f3}) are fulfilled. Then there exists $\delta >0$ such that for all $h\in L^\infty(\Omega)$ with $\|h\|_{L^\infty}<\delta$, problem (\ref{p1}) has at least one solution.
\end{theorem}

The proof of Theorem \ref{t1} combines the mountain pass theorem with an argument by contradiction. For a somewhat related result we refer to R\u adulescu \cite[Theorem~5.8]{radhindawi}.

The second objective in this paper concerns the existence of solutions to problem (\ref{p}) in the case where $\lambda\geq\lambda_1$, that is, provided that the operator $-\Delta-\lambda I$ is no longer coercive in $H^1_0(\Omega)$.

By multiplication with $\phi_1$ in (\ref{p}) we obtain
$$(\lambda_1-\lambda)\int_\Omega u\phi_1dx=\int_\Omega f(u)\phi_1dx.$$
This shows that problem (\ref{p}) cannot have {\it positive} solutions if $\lambda\geq\lambda_1$ and $f>0$ on $(0,\infty)$. That is why we impose the following additional hypothesis:
\begin{equation}\label{f5}
f:\RR\rightarrow\RR \mbox{ is strictly increasing and onto.}
\end{equation}

Under assumptions (\ref{f1})--(\ref{f3}) and (\ref{f5}), problem (\ref{p}) cannot have a {\it positive} solution. Indeed, by multiplication with $\phi_1$ in (\ref{p}), we obtain
$$(\lambda_1-\lambda)\int_\Omega u\phi_1dx=\int_\Omega f(u)\phi_1dx\,,$$
which is not possible if $u>0$ in $\Omega$. This shows that a natural question is to see if, under the same assumptions, problem (\ref{p}) admits a nontrivial {\it sign-changing} solution. The answer is {\it positive}, as shown by the next result.

\begin{theorem} \label{t2}
 Assume $\lambda\geq\lambda_1$ and hypotheses (\ref{f1})--(\ref{f3}) and (\ref{f5}) are fulfilled. Then problem (\ref{p}) has at least one nontrivial solution.
\end{theorem}

The proof of this result uses a dual variational method of Clarke and Ekeland \cite{cla1,cla2,eke}, which was initially introduced in the framework of Hamiltonian systems associated with a convex Hamiltonian. Toland \cite{toland} had introduced a related method for the study of variational problems involving the difference of two convex functions. We also refer to Brezis \cite{breams} for several applications of the dual variational principle to the qualitative analysis of nonlinear partial differential equations.

In the proofs of Theorems \ref{t1} and \ref{t2} we apply some methods developed in Brezis \cite{breams} and Brezis and Nirenberg \cite{brenir}

\section{Proof of Theorem \ref{t1}}
Arguing by contradiction, there is a sequence $(h_m)_{m\geq 1}$ in $L^\infty(\Omega)$ with $\|h_m\|_{L^\infty}\rightarrow 0$ such that for all $m\geq 1$, the problem
\begin{equation}\label{pm}
\left\{
\begin{array}{lll}
&-\Delta u_m-\lambda u_m = f(u_m)+h(u_m)\qquad &\mbox{in $\Omega$}\\
&u_m>0\qquad &\mbox{in $\Omega$}\\
 &u_m=0\qquad &\mbox{on $\partial\Omega$}
\end{array} \right.
\end{equation}
does not have any solution.

Set $F(t):=\int_0^tf(s)ds$. The energy functional associated to problem (\ref{pm}) is
$${\mathcal E}_m(u)=\frac 12\int_\Omega (|\nabla u|^2-\lambda u^2)dx-\int_\Omega F(u^+)dx
-\int_\Omega h_mudx\qquad\mbox{for all $u\in H^1_0(\Omega)$},$$
where $u^+:=\max\{u,0\}$.

We use in the proof the mountain pass theorem of Ambrosetti and Rabinowitz \cite{amra} in the following form.

\smallskip
{\bf Mountain Pass Theorem.} {\sl Assume $\Psi$ is a $C^1$ function on a Banach space $E$ and satisfies the Palais-Smale condition, that is,
$$\left\{\begin{array}{ll}
&\mbox{whenever a sequence $(v_m)$ in $E$ satisfies $|\Psi (v_m)|\leq C$ and $\Psi'(v_m)\rightarrow 0$ in $E^*$,}\\
&\mbox{there exists a subsequence of $(v_m)$ that converges in $E$.}\end{array}\right.$$
Assume also the following geometric hypotheses are fulfilled:
\begin{equation}\label{geo1}\left\{\begin{array}{ll}
&\mbox{there are constants $\delta>0$ and $c_0>0$ such that $\Psi (v)\geq c_0$}\\
&\mbox{for every $v\in E$ with $\|v\|=\delta$}\end{array}\right.\end{equation}
and
\begin{equation}\label{geo2}\Psi(0)<c_0\quad\mbox{and}\quad \Psi (v_0)<c_0\quad\mbox{for some $v_0\in E$ with $\|v_0\|>\delta$.} \end{equation}
Then there is a critical point $v$ of $\Psi$ such that $\Psi (v)\geq c_0$.
}

\smallskip
We check that ${\mathcal E}_m$ fulfills the geometric assumptions of the mountain pass theorem. We first observe that for all $u\in H^1_0(\Omega)$,
$${\mathcal E}_m(u)\geq\frac{\lambda_1-\lambda}{2\lambda_1}\,\|u\|^2-\int_\Omega F(u^+)dx-\|h_m\|_{L^\infty}\int_\Omega |u|dx.$$
Fix $\varepsilon>0$. Thus, by (\ref{f1}) and (\ref{f2}), there exists $C(\varepsilon)>0$ such that for all $t\in\RR$,
$$|f(t)|\leq\varepsilon\,|t|+C(\varepsilon)\, |t|^p,$$
hence
$$|F(t)|\leq\frac\varepsilon 2\, t^2+\frac{C(\varepsilon)}{p+1}\,|t|^{p+1}.$$
It follows that
$$\begin{array}{ll} {\mathcal E}_m(u)&\displaystyle\geq \frac{\lambda_1-\lambda}{2\lambda_1}\,\|u\|^2-\frac\varepsilon 2\int_\Omega (u^+)^2dx-
\frac{C(\varepsilon)}{p+1}\int_\Omega (u^+)^{p+1}dx-\|h_m\|_{L^\infty}\int_\Omega |u|dx\\
&\displaystyle\geq C_1\,\|u\|^2-C_2\varepsilon\,\|u\|^2-C_3(\varepsilon)\,\|u\|^{p+1}_{L^{p+1}}-C_4\,\|h_m\|_{L^\infty}
\,\|u\|\,.\end{array}$$
Thus, if $\|u\|=\delta>0$ is small enough, then
$${\mathcal E}_m(u)\geq c_0>0.$$
In the above argument we have also used Sobolev embeddings and the assumption $\|h_m\|_{L^\infty}\rightarrow 0$ as $m\rightarrow\infty$.

Next, for all $m\geq 1$,
$$\begin{array}{ll} {\mathcal E}_m(t\phi_1)&\displaystyle =
\frac{(\lambda_1-\lambda)t^2}{2}\int_\Omega\phi_1^2dx-\int_\Omega F(t\phi_1)dx-t\int_\Omega h_m\phi_1dx\\
 &\displaystyle \leq C_1t^2-\int_\Omega (C_2t^\mu\phi_1^\mu -C_3)dx-t\int_\Omega h_m\phi_1dx\\
 &\displaystyle =-C_2t^\mu\int_\Omega \phi_1^\mu dx+ C_1t^2 +C_3\,|\Omega|-O(t)\leq 0,
 \end{array}
 $$
if $t>0$ is large enough.

The verification of the Palais-Smale condition is standard for ${\mathcal E}_m$ and follows from the assumption $p<2^*-1$.

We now deduce from the mountain pass theorem that there exists $u_m\in H^1_0(\Omega)\setminus\{0\}$ such that ${\mathcal E}_m'(u_m)=0$. The corresponding critical value is ${\mathcal E}_m(u_m)=c_m\geq c_0$. 
Next, by multiplication with $u_m$ in (\ref{pm}) and using the assumption $\lambda<\lambda_1$ we deduce that
$\sup_m\|u_m\|<+\infty$. Standard elliptic regularity arguments imply that $(u_m)$ is bounded in $L^\infty(\Omega)$, hence in $L^p(\Omega)$ for all $1<p<\infty$. Schauder estimates (see Brezis \cite{brebook}) imply that $(u_m)$
is bounded in  $W^{2,p}(\Omega)$ for all $p<\infty$. By Sobolev embeddings, $(u_m)$ is bounded in $C^{1,\alpha}(\overline\Omega)$ for all $\alpha\in (0,1)$. Thus, up to a subsequence, $u_m\rightarrow u$ in $C^{1,\alpha}(\overline\Omega)$. Taking $m\rightarrow\infty$ we obtain
\begin{equation}\label{maxx}\left\{\begin{array}{lll}
&-\Delta u-\lambda u=f(u^+)\qquad &\mbox{in $\Omega$}\\
& u=0\qquad &\mbox{on $\partial\Omega$}.\end{array}\right.
\end{equation}
Moreover, the corresponding critical level is positive, that is,
$$\frac 12\int_\Omega (|\nabla u|^2-\lambda u^2)dx-\int_\Omega F(u^+)dx\geq c_0>0.$$
This shows that $u\not\equiv 0$. Applying now the maximum principle in (\ref{maxx}), we deduce that $u>0$ in $\Omega$. Since $u_m\rightarrow u$ in $C^{1,\alpha}(\overline\Omega)$, we obtain that $u_m>0$ in $\Omega$ provided that $m$ is large enough. Consequently, $u_m$ is a solution of problem (\ref{pm}) corresponding to $h=h_m$. This contradicts our assumption that problem (\ref{pm}) does not have any solution. The proof of Theorem \ref{t1} is now complete. \qed

\section{Proof of Theorem \ref{t2}}
The basic assumption $\lambda\geq\lambda_1$ implies that the geometric assumption (\ref{geo1}) in the statement of the mountain pass theorem is no longer fulfilled. That is why we use an idea found in the proof of Theorem~4 in Brezis \cite{breams}, which relies on the assumption that the nonlinear term $f$ is one-to-one and onto. We introduce a new unknown $v=f(u)$ and we prove that the energy functional associated to $v$ fulfills the hypotheses of the mountain pass theorem.

The solutions of problem (\ref{p}) correspond to the critical points of the energy functional
$${\mathcal E}(u):=\frac 12\int_\Omega (|\nabla u|^2-\lambda u^2)dx-\int_\Omega F(u)dx,\qquad u\in H^1_0(\Omega).$$

We first consider the case where the linear operator $(-\Delta -\lambda I)$ is invertible, hence $\lambda$ is not an eigenvalue of the Laplace operator. Thus, problem (\ref{p}) can be written, equivalently, $$u=(-\Delta-\lambda I)^{-1}f(u).$$
Set $T=(-\Delta-\lambda I)^{-1}$ and denote $f(u)=v$, hence $u=g(v)$ with $g=f^{-1}$.
Conditions (\ref{f2}) and (\ref{f4})  imply that $g\in L^r(\Omega)$, with $(N-2)/(N+2)<r<1$ if $N\geq 3$ and $0<r<1$ if $N\in\{1,\, 2\}$. Thus,  $T$ is a compact operator from $L^{r+1}(\Omega)$ into $L^{(r+1)'}(\Omega)$. This follows from the fact that $T$ maps $L^{r+1}(\Omega)$ into $W^{2,r+1}(\Omega)$ combined with the compact embedding
$W^{2,r+1}(\Omega)\subset L^{(r+1)'}(\Omega)$.

Using the dual formulation we have to find a nontrivial critical point for the functional
$${\mathcal J} (v):=\int_\Omega G(v)dx-\frac 12\int_\Omega vTvdx,\qquad G(y):=\int_0^yg(t)dt\,.$$
The above remarks show that ${\mathcal J}$ is well-defined and of class $C^1$ on the space $L^{r+1}(\Omega)$. Moreover, for all $v,\,\xi\in L^{r+1}(\Omega)$,
$${\mathcal J}'(v)(\xi)=\int_\Omega g(v)\xi dx-\int_\Omega\xi Tvdx.$$

We prove in what follows that ${\mathcal J}$ satisfies the hypotheses of the mountain pass theorem. We first observe that
$${\mathcal J}(v)\geq C_1\,\|v\|^{r+1}_{L^{r+1}}-C_2\, \|v\|^2_{L^1}\geq C_1\,\|v\|^{r+1}_{L^{r+1}}-C_3\, \|v\|^2_{L^{r+1}}\,.$$
Since $r+1<2$, this estimate shows that there are positive constants $R$ and $\rho$ such that ${\mathcal J}(v)\geq\rho$ for all $v\in L^{r+1}(\Omega)$ with $\|v\|_{L^{r+1}}=R$.
This shows that condition (\ref{geo1}) is fulfilled.

Let $\lambda_k>\lambda$ be an eigenvalue of $(-\Delta)$ and let $\varphi$ be a corresponding eigenfunction. Choose $v_0=t\varphi$, with $t>0$. Thus, since $r+1<2$,
$${\mathcal J}(v_0)\leq C_1\, t^{r+1}\int_\Omega |\varphi|^{r+1}dx+C_2-t^2(\lambda_k-\lambda)\int_\Omega |T\varphi |^2dx<0,$$
provided that $t>0$ is large enough. This implies that condition (\ref{geo2}) is fulfilled.

We prove in what follows that ${\mathcal J}$ satisfies the Palais--Smale compactness condition. Let $(v_n)\subset L^{r+1}(\Omega)$ be an arbitrary Palais--Smale sequence for
${\mathcal J}$. Since ${\mathcal J}'(v_n)\rightarrow 0$ in $L^{(r+1)'}(\Omega)$ we have
$$ g(v_n)-Tv_n=w_n\rightarrow 0\qquad\mbox{in } L^{(r+1)'}(\Omega).$$
Therefore
\begin{equation}\label{ps1}
\frac 12\int_\Omega g(v_n)v_ndx-\frac 12\int_\Omega v_n\, Tv_ndx=\frac 12\int_\Omega v_nw_ndx=o(1)\qquad\mbox{as $n\rightarrow\infty$}.\end{equation}
On the other hand, since $\sup_n|{\mathcal J}(v_n)|<+\infty$ we have
\begin{equation}\label{ps2}
 \int_\Omega G(v_n)dx-\frac 12\int_\Omega v_n\, Tv_ndx=O(1)\qquad\mbox{as $n\rightarrow\infty$.}
\end{equation}

Since $G(y)=yg(y)-F(g(y))$,
relations (\ref{ps1}) and (\ref{ps2}) yield
 $$ \int_\Omega \left[g(v_n)v_n-2F(g(v_n))\right]dx=O(1)\qquad\mbox{as $n\rightarrow\infty$.}$$
 Using now assumption (\ref{f3}) we deduce that
 $$ \int_\Omega g(v_n)v_n=O(1)\qquad\mbox{as $n\rightarrow\infty$.}$$
This implies that the sequence $(v_n)$ is bounded in $L^{r+1}(\Omega)$. Thus, up to a subsequence, $(v_n)$ converges weakly to $v$ in $L^{r+1}(\Omega)$. Using now the compactness of the operator $T$ we obtain $Tv_n\rightarrow Tv$ in $L^{(r+1)'}(\Omega)$, hence $v_n\rightarrow v$ in $L^{r+1}(\Omega)$. We conclude that ${\mathcal J}$ satisfies the Palais--Smale condition.

We now deduce from the mountain pass theorem that there exists $v\in L^{r+1}(\Omega)$, $v\not=0$, such that ${\mathcal J}'(v)=0$. Then $u=g(v)$ is a nontrivial solution of problem (\ref{p}). A standard bootstrap argument implies that $u\in H^1_0(\Omega)$.

\smallskip
It remains to treat the case where the operator $(-\Delta -\lambda I)$ is not invertible. This corresponds to a resonant problem, according to Landesman and Lazer \cite{lanlaz}.
We reduce this framework to the previous one and we try to put into evidence the same operator $T$ in a suitable function space.

Since $(-\Delta -\lambda I)$ is not invertible, then $\lambda$ is an eigenvalue of the Laplace operator. Let $E$ be the  finite dimensional space of eigenfunctions corresponding to $\lambda$. Then $H^1_0(\Omega)=E\oplus E^\perp$, where
$$E^\perp:=\left\{w\in H^1_0(\Omega);\ \int_\Omega w\xi dx=0\ \mbox{for all $\xi \in E$}\right\}.$$

We first observe that if $u$ is a solution of (\ref{p}) then for all $\xi\in E$,
$$\int_\Omega[(-\Delta u)\xi-\lambda u\xi]dx=\int_\Omega f(u)\xi dx.$$
Since $\xi\in E$ then $-\Delta \xi=\lambda\xi$ in $\Omega$. Thus, by Green's formula,
$\int_\Omega f(u)\xi dx=0$, hence $v=f(u)\in E^\perp$.
We now consider $T=(-\Delta-\lambda I)^{-1}$ as a linear continuous operator from $E^\perp$ into itself. This operator is well--defined and our problem (\ref{p}) becomes
\begin{equation}\label{newprob}
\left\{\begin{array}{ll}
& v\in E^\perp\\
&Tv-g(v)\in E.\end{array}\right.
\end{equation}

Conversely, we observe that if $v$ solves problem (\ref{newprob}) then $u=g(v)$ is a solution of (\ref{p}). This means that we can repeat the argument developed in the first part of the proof for the energy functional ${\mathcal J}$ but defined this time on $E^\perp$. This completes the proof of Theorem {\ref{t2}. \qed

\section*{Open problem.} We do not know if the result established in Theorem {\ref{t2} still remains true if assumption (\ref{f5}) is removed.

\section*{Further comments.}
The original proof of Ambrosetti and Rabinowitz \cite{amra} of the mountain pass theorem relies on some
deep deformation techniques developed by Palais and Smale \cite{Pa,palsma}, who put the main
ideas of the Morse theory into the framework of differential topology on
infinite dimensional manifolds. Brezis and Nirenberg provided in
\cite{brenir} a simpler proof which combines two major tools:
Ekeland's variational principle and the pseudo-gradient lemma.
Ekeland's variational principle is the nonlinear version of the
Bishop--Phelps theorem and it may be also viewed as a
generalization of Fermat's theorem.
Relevant extensions of the mountain pass theorem are due to Pucci and Serrin \cite{PS1,PS2,PS3} and Ghoussoub and Preiss \cite{ghpr}.
For applications of the mountain pass theorem to nonlinear partial differential equations we refer to Bonanno and Marano \cite{bonmar}, Carl and Motreanu \cite{carlmot}, Filippucci, Pucci and Robert \cite{fpr}, Krist\'aly, R\u adulescu and Varga \cite{krvbook}, and Marano and Motreanu \cite{marmot}. We also point the recent paper by Pucci and R\u adulescu \cite{puccirad} for a survey on the mountain pass theorem.


\section*{Acknowledgments.} The author is supported by Grant CNCSIS PCCE--8/2010
``Sisteme diferen\c{t}iale \^{\i}n analiza neliniar\u{a} \c{s}i aplica\c{t}ii".





\medskip
% The data information below will be filled by AIMS editorial staff
Received xxxx 20xx; revised xxxx 20xx.
\medskip

\end{document}
